Author's response to reviews

Title: Multiple Intracardiac Masses: Myxoma, Thrombus or Metastasis?

Authors:

Min-Ping Huang (MPHuang@cgmh.org.tw)
Morgan Fu (a9257@cgmh.org.tw)
Lee Wei-Chieh (leeweichieh@yahoo.com.tw)

Version: 2
Date: 25 February 2015

Author's response to reviews:

1. We reported a case about multiple intracardiac masses and learn how to differentiate diagnosis.
2. We reported a case about atypical picture of right atrial myxoma with two left ventricular thrombi.
3. We also proved intracardiac thrombi were resolved after optimal treatment.